

Lidija Klampfer, PhD

Southern Research Institute

Birmingham, AL

Dear Dr. Song

Attached is a revised version of our manuscript , entitled "Vitamin D and colon cancer".

One of the reviewers suggested to include a schematic representation of vitamin D action. We have provided new Figure 2, which illustrates that vitamin D inhibits Wnt signaling by acting on both tumor cells and the tumor-associated macrophages. We have indicated changes in the revised version in red.

Best regards,

Lidija Klampfer